Author's response to reviews

Title: E-health physical activity interventions and moderate-to-vigorous intensity physical activity levels among working-age women: a systematic review protocol

Authors:

Jennifer L Reed (jreed@ottawaheart.ca)
Stephanie A Prince (sprinceware@ottawaheart.ca)
Christie A Cole (cacole@ottawaheart.ca)
Kara A Nerenberg (kara.nerenberg@albertahealthservices.ca)
Swapnil Hiremath (shiremath@toh.on.ca)
Heather E Tulloch (HETulloch@ottawaheart.ca)
J George Fodor (george.fodor@gmail.com)
Agnieszka Szczotka (berkmanlibrary@ottawaheart.ca)
Lisa A. McDonnell (LMcDonnell@ottawaheart.ca)
Kerri-Anne Mullen (KMullen@ottawaheart.ca)
Andrew L. Pipe (apipe@ottawaheart.ca)
Robert D Reid (breid@ottawaheart.ca)

Version: 2 Date: 20 November 2014

Author’s response to reviews: see over
November 18, 2014

Dear Editor,

Please find enclosed a protocol article “E-health physical activity interventions and moderate-to-vigorous intensity physical activity levels among working-age women: a systematic review protocol” for your consideration for publication in *Systematic Reviews*. The subject of the paper is in line with your journal's interest related to all aspects of the design, conduct and reporting of systematic reviews. The proposed systematic review is novel and generates new knowledge in the area regarding effective e-health interventions for increasing physical activity levels of working-aged women in high-income/developed countries. Publishing the protocol and going through the peer review process will help to ensure its high quality. This protocol has not been funded and has therefore; not been subject to peer review.

The primary author of this manuscript is Jennifer Reed. This manuscript has not been published in another journal, book, or government publication, is not presently under consideration by another journal, and will not be submitted to another journal before a final editorial decision from *Systematic Reviews* is rendered. All previously published work that is cited in the manuscript has been fully acknowledged.

All of the authors have contributed substantially to this manuscript and have approved it for final submission. We have no competing interests to declare. Please find on the cover page a list (in order of authorship) and contact information for myself; the corresponding author.

Suggested peer reviewers for this manuscript include Dr. Robin J. Jacobs at the Nova Southeastern University, United States (email: rjacobs@nova.edu) and Dr. Colin Greaves at the University of Exeter, United Kingdom (email: Colin.Greaves@pms.ac.uk). We confirm that these peer reviewers have not published with any authors of the manuscript within the past five years, are not current collaborators and are not members of the same research institution.

Thank you for your consideration of this manuscript.

Sincerely,

Jennifer L. Reed, PhD, MEd CS